Author's response to reviews

Title: Pesticide exposure and risk of Parkinson's disease: a family-based case-control study

Authors:

Dana B Hancock (dana.hancock@duke.edu)
Eden R Martin (EMartin1@med.miami.edu)
Gregory M Mayhew (GMayhew@med.miami.edu)
Jeffrey M Stajich (stajich@chg.duhs.duke.edu)
Rita Jewett (r.jewett@miami.edu)
Mark A Stacy (mark.stacy@duke.edu)
Burton L Scott (scott007@mc.duke.edu)
Jeffery M Vance (jvance@med.miami.edu)
William K Scott (bscott@med.miami.edu)

Version: 3 Date: 5 March 2008

Author's response to reviews:

March 5, 2007

Melissa Norton, MD, Editor-in-Chief of the BMC series
BioMed Central Ltd.
Middlesex House, 34-42 Cleveland Street
London W1T 4LB, UK

Dear Dr. Norton,

We are resubmitting our manuscript entitled "Pesticide exposure and risk of Parkinson's disease: a family-based case-control study" with the requested formatting changes. More specifically, the tables have been renumbered to avoid the division of a table into parts a and b. References to the tables have been updated in the text. Also, website addresses have been moved from the text to the reference list.

Thank you in advance for your time. We look forward to seeing our manuscript in BMC Neurology.

Sincerely,

William K. Scott, PhD